Author's response to reviews

Title: Are classifications of proximal radius fractures reproducible?

Authors:

Fabio T Matsunaga (fteruo@gmail.com)
Marcel J S Tamaoki (marceltamaoki@terra.com.br)
Eduardo F Cordeiro (eduardo68@gmail.com)
Anderson Uehara (andersonuehara@hotmail.com)
Marcos H Ikawa (mhikawa@gmail.com)
Marcelo H Matsumoto (marcelohmatsumoto@terra.com.br)
Joao B G dos Santos (jbg santos@ig.com.br)
Joao C Belloti (jbellotij@terra.com.br)

Version: 4 Date: 28 September 2009

Author's response to reviews: see over
Dear Sir,

I am sending the second revision of my manuscript about proximal radius fractures classification.

According to the Associate Editor’s comments,

1) Job titles were removed
2) Reference 2 was updated
3) Tables were corrected
4) References were cited using brackets
5) Email address of each author were provided
6) Introduction was renamed to Background
7) Acknowledgements section was paced between About the authors and References

Please contact me if you have any questions about my manuscript.

With best wishes,

Fabio Teruo Matsunaga
Rua Borges de Lagoa, 783 — 5o andar
São Paulo (SP) — Brasil
CEP 04038-032